Author’s response to reviews

Title: Risk Indicators of Aggressive Periodontitis

Authors:
Khansa Ababneh (ktababneh@hotmail.com)
manal maslamani (mmaslamani@hsc.edu.kw)
Muna Abbadi (monlor22@yahoo.com)
anas taha (anass_taha@yahoo.com)
Jumana karasneh (jumana2003@yahoo.com)
amani sadi (amani.sadi@yahoo.com)
yousef khader (yousef.k@excite.com)

Version: 2 Date: 16 Jun 2019

Author’s response to reviews:

Dear Editor,

Thank you for your email and for giving us the chance to publish our research in this valuable journal. Also we like the reviewers we highly appreciate their valuable comments and giving their time to review our work.

This research was conducted at Jordan University of Science and Technology/ Jordan. It was not funded. Ethic approval was obtained from the The Ethic approval committee and consent form was signed by each patient participated in the study.

Each author contribution in the study as the following: KA and MM drafted the manuscript. KA is the principal investigator and designed the study. KA and MM developed the study protocol. KA, MA, AT, JK, AS were involved in the data collection as well. YK was involved in data analysis. All authors critically reviewed and revised the manuscript. The final version of the manuscript was approved by all authors.

All authors agree on publishing our research and BMC oral health journal and understand the need for publication fees.
Would you please inform us what is the next step to proceed with the publication?

Your help is highly appreciated.

Yours Sincerely

Manal Maslamani